Author's response to reviews


Authors:

Annick D Lenglet (annick@thelenglets.com)
Victoria Hernando (vhernando@isciii.es)
Pilar Rodrigo (mrodrigo@aragon.es)
Amparo Larrauri (alarrauri@isciii.es)
Juan DM Donado (jdonado@isciii.es)
Salvador de Mateo (smateo@isciii.es)

Version: 4 Date: 18 July 2007

Author's response to reviews:

MS: 9234843251203715

Madrid, 18 July, 2007

Dear Editor,

Thank you for sending us our manuscript entitled "Impact of flu on hospital based morbidity during 4 flu seasons in Spain, 2000-2004".

We have included a statement about ethical approval in the Methods section.

Thank you very much in advance.

On behalf of all the authors,
Victoria Hernando
National Centre for Epidemiology, Public Health Institute Carlos III
Tlf: +34 91 822 26 08
Fax: +34 91387 78 16
e-mail: vhernando@isciii.es